Author's response to reviews

Title: Inhibition of HIV-1 Entry by Extracts Derived from Traditional Chinese Medicinal Herbal Plants

Authors:

In-Woo Park (parki@iupui.edu)
Changri Han (cghan@163.com)
Xiaoping Song (xpsong@163.com)
Linden A Green (lhorten@indiana.edu)
Ting Wang (wangt@iupui.edu)
Ying Liu (yliu2@iupui.edu)
Changchun Cen (ccc369@136.com)
Xinming Song (xmsong@163.com)
Bioa Yang (bynag0612@163.com)
Guangying Chen (chgying23@163.com)
Johnny J He (jjhe@iupui.edu)

Version: 2 Date: 19 June 2009

Author's response to reviews:

Manuscript transferred from AIDS Research & Therapy. Submitted by Lisa Phelps on behalf of the authors.